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Mechanical work

RF coupler conditioning
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Changing CTS motors in CM14

UNIVERSITET ESS team done all work.

ESS team came to Freia by track with all necessary equipment.

Both stepper motors were replaced.

Limit switch on Cav2 side was replaced. ???
Tuner functionality was verified.

Leak check of insulation vacuum - OK.

Henry took 2 pieces of cryobellow protection with him to ESS




Progress from FREIA side

UNIVERSITET

CM status

Cryomodule installed back to the bunker.
Cryo, RF, vacuum, instrumentation
connection - Done.

Beam and insulation vacuum under the
pumping.

Bunker wall is under assembly.

FREIA cryo-plant status

He dryer filled by new absorber material
and regenerated.

Installed additional high pressure filter
between cold box and He dryer.
Starting liquefier - right now.
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& CTS homing test at warm W RPN,

Using Beckhoff driver. Go back by 1@ turns at a time until Low limit indicator goes on (limit switch

‘ I Sl open) on MTR1. Then it is not possible to continue negative.
Use the offset variable ESSPMAC:MTR1.0FF to start at @

O K Move negative

Turns|Limit switch (1 = closed, @ = open)
a8 1
-186 |1
-286 |1
-38 |1 Stops at -27.5 turns, -550@ steps limit switch indication on.

Positive direction

-27.5/8/1 Goes off after a while, -558@ steps
-2 |1
-1 |1
2] 1

Test the Homing procedure: OK (limit blinks)
Put the offset to -25.6

Using Beckhoff driver. Go back by 18 turns at a time until Low limit goes on on MTR2. Then it is
not possible to continue negative.
Use the offset variable ESSPMAC:MTR2.0FF to start at ©

Move negative

Turns|Limit switch (1 = closed, 8 = open)
e 1
-18 |1
-286 |1
-38@ |@ at -29.1 turns <=> -5828 steps

Positive direction

-29.1|@
-28 |1 on immediately at 29.1 turns <=» -5828 steps
-1e |1
e 1

Test the Homing procedure OK
Put the offset to -25.6




